Author's response to reviews

Title: Short term non-invasive ventilation post-surgery improves arterial blood-gases in obese subjects compared to supplemental oxygen delivery - a randomized controlled trial

Authors:

Martin Zoremba (zoremba@med.uni-marburg.de)
Gerald Kalmus (kalmus@med.uni-marburg.de)
Domenique Begemann (dominiquebegemann@gmx.net)
Leopold Eberhart (leberhart@web.de)
Norbert Zoremba (nzoremba@ukaachen.de)
Hinnerk Wulf (h.wulf@med.uni-marburg.de)
Frank Dette (dette@med.uni-marburg.de)

Version: 6 Date: 20 May 2011

Author's response to reviews:

Affilations were displayed as superscript numbers,
Abstract was pasted from the manuscript in the submission system.